Author’s response to reviews

Title: How does gender influence the recognition of cardiovascular risk and adherence to self-care recommendation? A study in Polish primary care

Authors:

Ireneusz Szymczyk (szymczykireneusz@poczta.onet.eu)
Ewa Wojtyna (ewa.wojtyna@us.edu.pl)
Witold Lukas (witold.lukas@onet.pl)
Joanna K&281;pa (joannaskorupka@interia.pl)
Teresa Pawlikowska (teresa.pawlikowska@warwick.ac.uk)

Version: 4 Date: 23 October 2013

Author’s response to reviews:

Dear Editors,

Please find attached a revised version of our manuscript “How does gender influence the recognition of cardiovascular risk and adherence to self-care recommendations? A study in Polish primary care”, which we would like to resubmit for publication as a research article in the BMC Family Practice.

The following page contains our point-by-point responses to each of the comments of the reviewer.

We hope that the revisions in the manuscript and our accompanying responses will be sufficient to make our manuscript suitable for publication in BMC Family Practice.

We look forward to hearing from you at your earliest convenience.

Yours sincerely,

Ewa Wojtyna, MD, PhD

Response to the comments of Reviewer (Kate Jolly)

1. I have two minor essential revisions:

   Please state when the questionnaire on cognitive appraisal was completed in phase 1 – was it immediately after the consultation with the doctor (i.e. during the same visit) or a few days later?

   Please add the dates over which the participants were recruited.

Response:

The suggested corrections have been made; the above mentioned information has been added to the text.